Author’s response to reviews

Title: Human insulin increases breast cancer risk in Taiwanese women with type 2 diabetes

Authors:

Chin-Hsiao Tseng (ccktsh@ms6.hinet.net)

Version: 2 Date: 2 December 2014

Author’s response to reviews:

The reuploaded manuscript has been modified according to the required formats.